# **RESEARCH ARTICLE**

Perceptions of the healthcare providers regarding acceptability and conduct of minimal invasive tissue sampling (MITS) to identify the cause of death in under-five deaths and stillbirths in North India: a qualitative study

Manoja Kumar Das<sup>1\*</sup>, Narendra Kumar Arora<sup>1</sup>, Reeta Rasaily<sup>2</sup>, Gurkirat Kaur<sup>1</sup>, Prikanksha Malik<sup>1</sup>, Mahisha Kumari<sup>1</sup>, Shipra Joshi<sup>1</sup>, Harish Chellani<sup>3</sup>, Harsha Gaekwad<sup>4</sup>, Pradeep Debata<sup>3</sup> and K. R. Meena<sup>3</sup>

# Abstract

**Background:** India contributes the highest share of under-five and neonatal deaths and stillbirths globally. Diagnostic autopsy, although useful for cause of death identification, have limited acceptance. Minimally invasive tissue sampling (MITS) is an alternative to autopsy for identification of the cause of death (CoD). A formative research linked to pilot MITS implementation was conducted to document the perceptions and attitudes of the healthcare professionals and the barriers for implementation.

**Methods:** This exploratory qualitative study conducted at a tertiary care hospital in New Delhi, India included the hospital staffs. In-depth interviews were conducted with the doctors, nurses and support staffs from pediatrics, neonatology, obstetrics and forensic medicine departments. Inductive data analysis was done to identify the emerging themes and codes.

**Results:** A total of 26 interviews (doctors, n = 10; nurses, n = 9 and support staffs, n = 7) were conducted. Almost all professional and support staffs were positive about the MITS and its advantage for CoD identification including co-existing and underlying illnesses. Some opined conduct of MITS for the cases without clear diagnosis. All participants perceived that MITS would be acceptable for parents due to the non-disfigurement and preferred by those who had unexplained child deaths or stillbirths in past. The key factors for MITS acceptance were appropriate communication, trust building, involvement of senior doctors, and engagement of the counselor prior to deaths and training of the personnel. For implementation and sustenance of MITS, involvement of the institute authority and government stakeholders would be essential.

(Continued on next page)

\* Correspondence: manoj@inclentrust.org <sup>1</sup>The INCLEN Trust International, New Delhi 110020, India Full list of author information is available at the end of the article





# **Open Access**

© The Author(s). 2020 **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

### (Continued from previous page)

**Conclusions:** MITS was acceptable for the doctors, nurses and support staffs and critical for better identification of the causes of death and stillbirths. The key facilitating factors and challenges for implementing MITS at the hospital in Indian context were identified. It emphasized on appropriate skill building, counseling, system organization and buy-in from institution and health authorities for sustenance of MITS.

**Keywords:** Minimal invasive tissue sampling, Cause of death, Child, Neonates, Stillbirth, Healthcare providers, Hospital, Acceptance, India

# **Plain English summary**

Despite significant progress in child and maternal health indicators, India contributes the largest share of childhood deaths and stillbirths globally. For designing and implementing effective clinical and public health programs, adequate knowledge about the causes childhood deaths and stillbirths are essential. The currently available causes of death (CoD) and stillbirth are primarily derived from verbal autopsy (VA), a history and questionnaire based enquiry which has several limitations. Complete diagnostic autopsy is the best possible method for CoD identification, but with limited acceptability. Minimally invasive tissue sampling (MITS) technique has recently emerged as an alternative to autopsy. Before implementation, it was necessary to understand the perceptions, practices, acceptance and barriers among the healthcare professionals regarding MITS. An exploratory qualitative study was undertaken at a tertiary care hospital in New Delhi, India using in-depth interviews with different cadres of healthcare professionals from paediatrics, neonatology, obstetrics and forensic medicine departments. MITS was acceptable for almost all healthcare professionals and support staffs. They perceived MITS to be useful for identification of underlying causes of death and stillbirth, especially in cases of unexplained deaths or stillbirths. They perceived appropriate communication, trust building and involvement of senior doctors as the key factors for MITS acceptance by parents.

# Background

Globally, about 5.3 million under-five children and 2.6 million neonates died globally in 2018 [1]. About 2.6 million stillbirths also occur every year globally [2–4]. India was the largest contributor of under-five and neonatal deaths and stillbirths in 2018 [1]. About 20% of the global stillbirths occured in India [5]. Despite the significant progress in child and maternal health indicators for India, the decline in neonatal death and stillbirth rates are slower. Effective implementation of cost-effective interventions at scale with quality can prevent majority these childhood deaths [6]. As per National Health Policy (2017), India targets reducing the under-five and neonatal mortality rates to 23 and < 10 per 1000

livebirths by 2025 [7]. For designing and implementing effective interventions, adequate knowledge about the infectious and non-infectious causes of death (CoD) and stillbirths are critical.

The current information on the CoD in children is primarily based on verbal autopsy (VA), as many children die at community level or before reaching hospitals or any healthcare provider. Out of those who reach health facilities, the documentation of CoD is not adequate due to inadequate investigation and incomplete information about CoD in death declaration, often missing the coexisting illnesses or factors [8–12]. Complete diagnostic autopsy, although can provide critical information on CoD, its conduct is dismal for non-legal cases due to the sociocultural and religious norms and technical and infrastructural challenges [13–17]. The CoDs identified through VA have several limitations due to the non-specific clinical features in newborns and infants, inability to capture the coexisting and underlying problems [18-21]. VA is not much useful in identification of the cause(s) of stillbirths. The minimally invasive tissue sampling (MITS) technique involves detailed physical examination, imaging, targeted tissue sampling and processing for histopathologic, microbiological and other desired investigations. MITS has been observed to be efficient for CoD identification [22-24]. MITS is less invasive, non-disfiguring and time and cost efficient, but has results comparable to the standard autopsy for CoD [25-27]. MITS is being used in project mode in few countries. The formative researchers and early implementation experiences have revealed positive theoretical acceptability for MITS among parents, community members and religious leaders [28-32]. Acceptance for MITS among the healthcare providers at facility and community levels have been encouraging [33-36]. The acceptance of MITS among the families and community may vary across the countries due to the sociocultural and religious contexts. The acceptance among the healthcare providers may also vary based on the experience, infrastructure, and sociocultural norms.

A pilot project was undertaken in India as part of the Child Health and Mortality Prevention Surveillance Network, to document the feasibility, acceptability, capacity building and standardization to determine the CoD. This pilot project included conduct of MITS to identify the CODs for under-five deaths including neonatal deaths and stillbirths. The healthcare professionals and hospital staffs interact closely with the parents and family members during the treatment and death declaration and are likely to influence the MITS acceptability. Thus, for conduct of MITS, it was important to understand the perception and attitudes of these key stakeholders. A formative research was conducted to understand the perceptions, practices, contextual factors and barriers for acceptance and conduct of MITS. The study protocol has been published earlier [37].

# Methods

# Study design and setting

This exploratory qualitative research adopted grounded theory technique. This study was conducted at Safdarjung Hospital (SJH), New Delhi, India, during December 2018 to January 2019.

## Study participants

The healthcare providers from paediatrics (neonatology and children wards), obstetrics and the support staffs from SJH were purposively sampled. The healthcare providers included doctors (junior residents, senior residents and faculty) and nurses from different wards/units. The faculty and mortuary staff from forensic medicine department were also included.

# Data collection

Guides for IDIs were prepared covering the domains and issues to be explored based on the literature review (Supplementary document). The IDI guides had open-ended questions that explored the perceptions regarding the cause(s) of death in children or stillbirth, attitudes and opinions about the conduct, benefit, acceptance and challenges for MITS. During the interview, before asking about MITS, the interviewer described the MITS procedure components. A team of investigator and trained research staffs conducted the interviews at places convenient for the participants. During the interview, free flow of the information was encouraged with use of probes and additional issues were discussed, as they emerged. The IDIs were conducted in either local language (Hindi) or English as convenient for the participant and audio-recorded with consent. The research team took detailed field notes and noted the verbal and non-verbal languages and expressions. The interviews usually lasted for about 45-60 min.

# Data handling and analysis

The audio records and field notes were transcribed verbatim and then translated into English, if needed. An independent member checked for the completeness and correctness of the transcriptions and quality of translation with the audio-records and field notes. The data was entered using INCLEN Qualitative Data Analysis Software (IQDAS), which allows data entry, organization and retrieval for analysis in Indian languages and English. The entered data were checked completeness and correctness. The data were saved into the server and daily back up was taken. The inductive data analysis followed the steps: free listing, domain identification, coding, and cross tabulation. The emerged codes and themes were discussed periodically and the discrepancies were resolved. From the codes, the axial codes were identified followed by grouping into selective codes and organized under key themes. An iterative process was adopted for the transcript reading, coding and theme identification till consensus was reached. The themes across the participant categories were triangulated for consistencies and diversions. The findings were summarized using semi-quantitative qualifiers: very few (< 10%), some (10-24%), about half (25-49%), majority (50-75%), most (76-89%) and almost all (>90%).

#### **Ethical considerations**

Ethical approval for the study was obtained from the participating institute ethics committees (INCLEN Ethics Committees, Ref: IIEC 51 and Institutional Ethics Committee, VMMC and Safdarjung Hospital, Ref: IEC/SJH/VMMC/Project/August-2017/1000). The study participants were recruited after obtaining written informed consent including permission for audio recording and use of anonymized quotes. The interviews were conducted in isolation and no one other than the participant and interviewer was present during the process. Confidentiality and anonymity of participants were assured.

# Results

A total of 26 IDIs were conducted with doctor (n = 10), nurses (n = 9) and support staffs (n = 7) (Table 1). All the participants approached consented and participated. Based on the data analysis, the key thematic areas identified were: (1) perceived benefit; (2) acceptance by families; (3) hospital and system readiness and (4) factors influencing implementation. The findings are presented below.

# Perceived benefit

#### Better and accurate diagnostic ability

Most healthcare professionals and support staffs favoured MITS considering that it might offer better and accurate diagnosis. The paediatricians expressed that in several cases they were unable to make conclusive diagnosis or CoD due to shorter time of hospitalisation or critical condition and limited opportunity for the investigations. The obstetricians mentioned that causes of stillbirths remain unknown in most scenarios. The detailed examination, imaging, body fluids and tissue samples collected by needle biopsies would potentially allow better diagnosis. The microbiological, pathological and molecular diagnostic tests could provide valuable information for determining the causes of death and stillbirth. The forensic expert

# Table 1 The stakeholders participated in the study

SI	Stakeholder category	Number
1	Profession	
1.1	Doctors, n	10
	- Pediatrics (Junior resident, Senior resident and Faculty, 1 each)	3
	- Neonatology (Junior resident, Senior resident and Faculty, 1 each)	3
	- Obstetrics (Junior resident, Senior resident and Faculty, 1 each)	3
	- Forensic medicine (Faculty)	1
1.2	Nurses, n	9
	- Pediatrics (from different wards)	3
	- Neonatology	3
	- Obstetrics (labour rooms and postnatal ward)	3
1.3	Support staffs, n	7
	- Pediatrics (from different wards)	2
	- Neonatology	2
	- Obstetrics (labour rooms and postnatal ward)	2
	- Mortuary	1
2	Gender, n	26
	- Female	19
	- Male	7
3	Age group, median (range) in years	
	- Faculty members	47 (45–52)
	- Resident doctors	28 (26–29)
	- Nurses	45 (37–52)
	- Support staffs	42 (34–48)
4	Service tenure, median (range) in years	
	- Faculty members	15 (13–20)
	- Residents	3 (2–5)
	- Nurses	16 (13–22)
	- Support staffs	15 (12–18)

indicated that the acceptance for autopsy was very low and in many medicolegal cases arguments happen with the families over the body disfigurement and time taken.

# Identification of less common and co-existing illnesses

Doctors perceived that MITS has potential to allow identification of less common, coexisting and underlying illnesses. The paediatricians opined that it would be very useful in neonatal deaths, where sepsis is assigned as CoD in most cases. All informants felt that MITS would be useful to improve the knowledge and input for the treatment and programs.

# General or selective approach

The professionals had different opinions on case selection for MITS. While majority preferred conduct of MITS in all cases where the diagnosis or CoD remains unknown, some felt that the cases be selected by the treating doctors, as per the need. Few doctors perceived MITS to be futile in cases with clear diagnosis or with major underlying disease or malformation.

# Potential acceptance by parents Acceptance of MITS by parents

The health professionals and staffs indicated that MITS in view of the no disfigurement and organ retrieval, MITS could be acceptable for the families. Most of them perceived that the willingness for MITS may depend on the perceived benefit by the families. They felt that the families with past child death or bad obstetric history might be more willing for MITS than others. Majority of the informants also felt that the education, sociocultural and religious factors would influence the acceptance for MITS. Few doctors felt that the MITS findings might be counterproductive and trigger aggression in some cases, where the family don't accept the death or agree with the CoD.

#### When to approach for consent

Almost all opined that the family may be approached for consent after few minutes of death declaration, as some time may be needed by them to accept the death and come to a state of potential understanding. Regarding the person to be approached for consent, almost all were unanimous that the father, grandfather or other male member present in the hospital should be approached. According to the obstetricians and nurses, the stillbirths were usually declared to the husband or family members and not the woman directly. The nurse and support staffs felt that the parents and family who lost their child or pregnancy would be in a grief state and it would be challenging to approach them at that time.

"Most of the time, they will not give consent, because you can imagine that most of the doctor's face violence and verbal abuse, imagine if you will say that we are going to take samples out of body then, maybe that will counter benefit than beneficial. Then they will ask that what is benefit of that, they are not getting any benefit, maybe hospital is getting any benefit, and their baby has died." (Doctor-Pediatrics)

"It (MITS) will help, but I don't think so people will give permission to that. It will depend on, I will tell you, who accept death and who don't." (Doctor-Pediatrics)

# Hospital and system readiness Readiness of healthcare professionals

Almost all of the participants mentioned that the hospital was already overloaded and the professionals were overstretched with the existing workload. According to them the MITS would be a time taking procedure considering the counselling, consent and conduct of the procedure. According to them, separate teams would be needed for obtaining the consent and conduct of MITS. Additionally the other departments and teams like microbiology, pathology and radiology should be involved from the preparatory phase. They perceived that if it has to be conducted in a project mode, it might be feasible, but for routine services, additional planning would be needed.

"The workload will increase, if the present doctors have to perform these tests. It would be better if other team is allotted to perform MITS procedure." (Doctor-Pediatrics)

"The workload on doctors will increase. Here staff nurses don't do any sampling. Occasionally put the IV cannula. All invasive procedures are done by doctors. By this the pressure on doctors will increase." (Nurse-Pediatrics)

# Skills and training

The health professionals opined that appropriate training of the doctors, nurses and counsellors shall be needed for obtaining consent and conduct of MITS. They felt that the skills of the counselling team would be critical to interact with families for obtaining consent. The doctors opined that appropriate orientation and protocol needs to be prepared for all the involved procedures and departments for appropriate function and outcome. Almost all of the participants expressed that the declaration of stillbirth and death affected them and their emotions, although they have learnt over time how to handle it. They opined that there should be training sessions for young doctors and nurses on how to deal with the deaths and related counselling.

"We don't get any specific teaching and reading materials on the part of counseling. It is self-reading, observational, sometimes told in classrooms." (Doctor-Pediatrics)

"It is very bad for us, no matter how rigid we appear on the face, but in the night when we go home, there is a striking force that child expired on our duty. Many times I am not able to sleep because of this fact." (Doctor-Pediatrics)

# Infrastructure, equipment and cost implications

The doctors stressed on appropriate infrastructure creation, equipment and supplies for conduct of MITS, which should be funded by government or from project, as appropriate. The participants opined that the MITS procedure should be free of cost for the family, as additional cost would be a challenge for obtaining consent. The nurses and support staffs felt that a separate area for MITS be prepared, which could be accessed easily and allow waiting of parents and family members. Most of the doctors felt that if the institute leadership and government decision makers are convinced, cost would not be a concern. According to them, if huge investments are needed, then it might pose challenge for sustaining beyond the project period.

# Factors influencing implementation *Effective communication and trust*

The participants were unanimous about the need for transparent and frequent communication with the parents to build trust and confidence. According to most of the participants, the counsellors should preferably engage with the families before the death to provide some time for rapport and confidence building. The quality and ability of the counsellor was indicated as critical by majority of the doctors and nurses. They also felt that senior doctor's presence could improve acceptance and consent by families. Some of the nurses and support staffs felt that after conduct of the MITS, the family should be shown the body before handover. Few participants opined that training would be needed for all the staffs on appropriate communication with the patients and families for effective implementation.

"No, I think it is important. I have not taken any training in this field, so I sometimes explain them. We don't know the proper way. Whatever way I explain they listen to it, they don't react abnormally, but still I think it can be improved." (Doctor-Obstetrics)

#### Conduct of the MITS by trained person

The doctors felt that in view of the importance and value, a trained person should conduct MITS and should not be taken for granted. Some of the professionals were apprehensive that MITS, if conducted by the junior residents might be useless.

"Availability of trained person and space is challenge. We don't have any trained person, we don't know who is going to do it and where they are going to do it. You do it and then put the things in reverse gear." (Doctor –Obstetrics)

#### Timing for counselling and conduct of MITS

Almost all of the participants highlighted that MITS should be conducted as quickly as possible to minimise the denials. Most of them were unsure about the appropriate timing to approach for consent and felt that the team on ground may decide the timing. Almost all of the participants opined that the families would like to take the body and leave as soon as possible after the death. Some felt that once the family members appear settled and in a state to discuss, they should be approached for consent. Some of the respondents felt that the deaths occurring towards the later part of the day or at night might be challenging, as the families would like to move quickly and or might not have the decision maker present. The health professionals mentioned that a quicker conduct of MITS, within 1-2hwould be acceptable for the families and delay in procedure would reduce acceptance.

# Sharing the findings with parents and family

The doctors and nurses felt that the findings of MITS should be shared with the parents and family at the earliest possible. According to them, if delayed, it might not be considered useful for them. Also they felt that the treating senior doctor should disclose and counsel the parents.

# Institute and government facilitation

The doctors were of opinion that for effective implementation, there is need for buy-in and facilitation from the institute leadership and government to ensure funding for the infrastructure, manpower and supporting cost. Additionally for keeping the relevant stakeholders and departments engaged in the implementation, the support and facilitation from higher officials shall be critical.

# Discussion

This is the first study from India to our best of knowledge documenting the perceptions and attitudes of the health professionals and support staffs on MITS and related issues. When autopsy is not acceptable to the families, the MITS is an alternative to know causes of death and stillbirth. But for conduct and sustenance the MITS within the existing system, the view of healthcare professionals and support staffs are important. We found that the hospital staffs were positive about acceptance of MITS among the parents, if offered to know the CoD in children and stillbirths. According to them, tThe factors in favor of MITS acceptance were no disfigurement and organ removal. They opined that the acceptance would be more among the parents with past experience of child death or bad obstetric history. These factors were similar to those identified in studies from United Kingdom, Pakistan, Gabon, Kenya, Mali, Mozambique and Kenya [13, 33–35, 38].

The study explored the factors that might influence implementation of MITS in a tertiary care hospital in India. Overall, the professionals and staffs expressed MITS to have value addition for identification of causes of death and stillbirth, which shall improve the treatment and health programs. Higher possibility of identification of underlying and uncommon causes with better diagnostic accuracy were considered as the benefits of MITS, which were similar to the other studies [34, 35, 39].

A formative research in Pakistan observed possible high theoretical MITS acceptance among the parents, especially by those who have experienced multiple neonatal deaths, stillbirths or miscarriages. The trust on healthcare providers and hospital, effective counseling and promptness in conducting MITS were identified as the influencing factors for the acceptance. The participants also opined that appropriate healthcare providers should be identified for training in MITS to retain interest [34]. These observations were similar to the findings from our study.

Healthcare providers from Mozambique had negative opinion about complete autopsy, as it was not easy to perform, procedures involved, unpleasantness, physical and mental discomfort for the performer and required specialized training and competency to conduct. Those who had experience of conducting autopsy found the minimally invasive autopsy (MIA) as simple, easy and quick technique that could be done anywhere and by anyone after appropriate training even in the absence of formal pathology facilities. They also found the MIA to be cleaner, less messy, without unpleasant smell and higher family compliance. But some healthcare providers felt that the yield and productivity would be inferior to the complete autopsy [35]. The findings from our study had similar observations.

A survey among healthcare providers in United Kingdom reported higher acceptability for MIA compared to autopsy across the ethnic and religious group. Some of the respondents had concerns about the accuracy of the MIA. Some of the healthcare providers had concern about the acceptance among Indian, Asian or Arab ethnicity in view of their religious and cultural practices [40].

However some professionals felt that MITS should be reserved for patients with no or incomplete diagnosis. These observations were similar to the observations from other studies [34, 35].

Several studies have identified the challenges for obtaining consent for postmortem examination and procedures for children [13, 32, 41, 42]. A review of 34 reports on acceptance of autopsy and MIA identified seven themes describing the barriers and six facilitating themes for postmortem acceptance among parents. The seven barrier themes were dislike of invasiveness, practicalities of the procedure, organ retention issues, protective parenting, communication and understanding, religion and culture and professional or organisational factors. The six facilitating themes for consent included desire for information, contribution to research, coping and well-being, respectful care, minimally invasive procedure, and policy and practice [13]. The participants mentioned the challenges related to acceptance of postmortem by the families including their willingness to know the CoD, disfigurement, delay in the process and final rituals and religious norms. The participants perceived the advantages and lesser concerns compared to complete autopsy would facilitate its acceptance by the families.

To encourage the acceptance of MITS, the participants indicated the importance of appropriate communication, effective counselling and explanation by the health professionals. The study found that the counselors and health professionals, especially the younger doctors and nurses, should be trained for effective communication and counselling. It was also found that the counselors should be involved prior to the death along with the treating team to build rapport. The study identified that counseling by the senior doctors and the staffs with good rapport with the parents would improve the chance of acceptance. These findings were similar to the other studies on MITS and postmortem autopsy [34, 41, 42]. Although no definite time window to approach for consent was identified, the participants felt that it should be decided by the counselor and doctor. It was found that the MITS should be conducted quickly to minimize logistic hassles for families. The participants suggested early feedback to the parents on findings for appropriate action and retain the confidence on hospital.

For implementation of MITS, several system level challenges were identified including manpower shortage, additional work for the existing staffs, infrastructural and equipment support, cooperation from the concerned departments for processing the samples and cost. It also found that for sustenance beyond the project period, engagement of the institute authority and government decision makers are needed.

The opinions from different categories of healthcare providers from the concerned departments was the strength. The study had some limitations. The results represent the views and attitudes of the healthcare professionals and staffs from one hospital in India, which may differ from the opinion of health professionals from other parts of the country. The MITS acceptability explored was hypothetical, prior to the implementation, which may change with real experience. Future research may be undertaken to document the changes in the views and attitudes of the healthcare professionals with implementation of MITS including the facilitators and barriers.

# Conclusion

Overall, MITS was viewed positively and acceptable to the healthcare professionals and staffs at the tertiary care hospital. The healthcare providers perceived MITS to be acceptable to the parents and assist in better diagnosis and appropriate treatment. The study identified several concerns and challenges for the professionals, health system and administration for effective implementation and sustenance in Indian context. The study underscored the importance of appropriate communication, trust building and counselling for MITS acceptance by the parents. Appropriate manpower engagement, training, infrastructural support and administrative facilitation would help successful implementation and sustenance of MITS.

## Supplementary information

Supplementary information accompanies this paper at https://doi.org/10. 1186/s12913-020-05693-6.

Additional file 1.

#### Abbreviations

CoD: Causes of death; IDI: In-depth interview; IQDAS: INCLEN Qualitative Data Analysis Software; MITS: Minimally invasive tissue sampling;

NICU: Newborn intensive care unit; SJH: Safdarjung Hospital; SNCU: Sick newborn care unit; VA: Verbal autopsy

#### Acknowledgements

We acknowledge the participation of the parents and community members for their contribution. We appreciate the participation and support from paediatricians, obstetricians, residents, nurses and record section officials of . VMMC and Safdarjung Hospital, New Delhi in conduct of this study. We highly value the guidance from the Technical Advisory Group members, Dr. Siddarth Ramii, Maulana Azad Medical College, New Delhi; Dr. Gagandeep Kang, Translational Health Science and Technology Institute, Faridabad, Haryana; Dr. Sunita Saxena, National Institute of Pathology, New Delhi; and Dr. Yogesh Jain, Jan Swasthya Sahyog, Bilaspur, Chattisgarh. We acknowledge the cooperation from the co-investigators: Dr. Usha Agrawal and Dr. Fauzia Siraj, National Institute of Pathology (Indian Council of Medical research), New Delhi; Dr. Pratima Mittal, Dr. Rajni Gaind, Dr. K. C. Agarwal, Dr. Archana Kashyap and Dr. Manisha, Safdarjung Hospital and Vardhman Mahavir Medical College, New Delhi. We also acknowledge the support from other INCLEN team members including Deepak Singh, Vinod Kumar, Chandan Singh, Amit Kumar, Bablu and Rajender. We appreciate the technical assistance received from CHAMPS project team members.

#### Disclosure statement

None. There is no financial interest or benefit for the authors arisen from this project or its direct application.

#### Authors' contributions

The qualitative study was conceptualized by MKD and NKA. MKD, GK, PM, MK, and SJ collected and analyzed the data. HC, HG, PD, KRM and RR facilitated the data collection. MKD wrote the first draft of the manuscript. All authors reviewed, provided critical input and approved the final version. The content represents the views of the authors alone and do not necessarily represent the official positions of their organizations.

#### Funding

This study is funded to The INCLEN Trust International by Bill and Melinda Gates Foundation (OPP1184205) through Indian Council of Medical Research (no 5/7/1504/2016-CH). The funders had no role in the study design, data collection, analysis, interpretation, decision to publish or preparation of the manuscript.

#### Availability of data and materials

The datasets used and analyzed during the current study are available from the corresponding author on reasonable request.

#### Ethics approval and consent to participate

The study protocol was reviewed and approved by Institute Ethics Committees of The INCLEN Trust International (Ref: IIEC 51) and VMMC and Safdarjung Hospital (Ref: IEC/SJH/VMMC/Project/August-2017/1000). The indepth interviews and focus group discussions with stakeholders were done after obtaining written informed consent.

#### Consent for publication

Not applicable.

#### **Competing interests**

The authors declare that there is no competing interests and conflict of interest. The author, RR, a staff members at Indian Council of Medical Research, who coordinated between different institutions for implementation the study and was not involved in the implementation or interpretation of the results.

#### Author details

<sup>1</sup>The INCLEN Trust International, New Delhi 110020, India. <sup>2</sup>Division of Reproductive Biology Maternal and Child Health, Indian Council of Medical Research, New Delhi 110029, India. <sup>3</sup>Department of Pediatrics, Safdarjung Hospital and Vardhman Mahavir Medical College, New Delhi 110029, India. <sup>4</sup>Department of Obstetrics and Gynaecology, Safdarjung Hospital and Vardhman Mahavir Medical College, New Delhi 110029, India.

## Received: 21 July 2020 Accepted: 31 August 2020 Published online: 04 September 2020

#### References

- United Nations Inter-agency Group for Child Mortality Estimation (UN. Levels & Trends in Child Mortality: Report 2019, Estimates developed by the United Nations Inter-agency Group for Child Mortality Estimation. United Nations Children's Fund, New York; 2019 [cited 2019 Dec 30]. Available from: https:// www.unicef.org/media/60561/file/UN-IGME-child-mortality-report-2019.pdf.
- Reinebrant H, Leisher S, Coory M, Henry S, Wojcieszek A, Gardener G, et al. Making stillbirths visible: a systematic review of globally reported causes of stillbirth. BJOG Int J Obstet Gynaecol. 2018;125(2):212–24.
- Lawn JE, Blencowe H, Waiswa P, Amouzou A, Mathers C, Hogan D, et al. Stillbirths: rates, risk factors, and acceleration towards 2030. Lancet Lond Engl. 2016;387(10018):587–603.
- Saleem S, Tikmani SS, McClure EM, Moore JL, Azam SI, Dhaded SM, et al. Trends and determinants of stillbirth in developing countries: results from the global Network's population-based birth registry. Reprod Health. 2018;15(Suppl 1):100.
- Blencowe H, Cousens S, Jassir FB, Say L, Chou D, Mathers C, et al. National, regional, and worldwide estimates of stillbirth rates in 2015, with trends from 2000: a systematic analysis. Lancet Glob Health. 2016;4(2):e98–108.
- United Nations Inter-agency Group for Child Mortality Estimation (UN. Levels & Trends in Child Mortality: Report 2018, Estimates developed by the United Nations Inter-agency Group for Child Mortality Estimation. United Nations Children's Fund, New York; 2018 [cited 2019 Jun 28]. Available from: https://childmortality.org/files\_v22/download/UN%20IGME%20Child%2 0Mortality%20Report%202018.pdf.
- Ministry of Health and Family Welfare. National Health Policy 2017 [Internet]. Government of India; 2017 [cited 2018 Sep 25]. Available from: https:// mohfw.gov.in/sites/default/files/9147562941489753121.pdf.
- Mathers CD, Fat DM, Inoue M, Rao C, Lopez AD. Counting the dead and what they died from: an assessment of the global status of cause of death data. Bull World Health Organ. 2005;83(3):171–7.
- Deshmukh V, Lahariya C, Krishnamurthy S, Das M, Pandey R, Arora N. Taken to health care provider or not, under-five children die of preventable causes: findings from cross-sectional survey and social autopsy in rural India. Indian J Community Med. 2016;41(2):108.
- Gupta N, Bharti B, Singhi S, Kumar P, Thakur JS. Errors in filling WHO death certificate in children: lessons from 1251 death certificates. J Trop Pediatr. 2014;60(1):74–8.
- Hernández B, Ramírez-Villalobos D, Romero M, Gómez S, Atkinson C, Lozano R. Assessing quality of medical death certification: Concordance between gold standard diagnosis and underlying cause of death in selected Mexican hospitals. Popul Health Metr. 2011 [cited 2020 Apr 17];9(1). Available from: http://pophealthmetrics.biomedcentral.com/articles/10.1186/1478-7954-9-38.
- Hazard RH, Chowdhury HR, Adair T, Ansar A, Quaiyum Rahman AM, Alam S, et al. The quality of medical death certification of cause of death in hospitals in rural Bangladesh: impact of introducing the International Form of Medical Certificate of Cause of Death. BMC Health Serv Res. 2017 Dec [cited 2020 Apr 17];17(1). Available from: http://bmchealthservres. biomedcentral.com/articles/10.1186/s12913-017-2628-y.
- Lewis C, Hill M, Arthurs O, Hutchinson C, Chitty L, Sebire N. Factors affecting uptake of postmortem examination in the prenatal, perinatal and paediatric setting. BJOG Int J Obstet Gynaecol. 2018;125(2):172–81.
- Lishimpi K, Chintu C, Lucas S, Mudenda V, Kaluwaji J, Story A, et al. Necropsies in African children: consent dilemmas for parents and guardians. Arch Dis Child. 2001;84(6):463–7.
- Ugiagbe EE, Osifo OD. Postmortem examinations on deceased neonates: a rarely utilized procedure in an African referral center. Pediatr Dev Pathol Off J Soc Pediatr Pathol Paediatr Pathol Soc. 2012;15(1):1–4.
- Fligner CL, Murray J, Roberts DJ. Synergism of verbal autopsy and diagnostic pathology autopsy for improved accuracy of mortality data. Popul Health Metrics. 2011;9:25.
- Turner GDH, Bunthi C, Wonodi CB, Morpeth SC, Molyneux CS, Zaki SR, et al. The Role of Postmortem Studies in Pneumonia Etiology Research. Clin Infect Dis. 2012;54(suppl\_2):S165–71.
- Soleman N, Chandramohan D, Shibuya K. Verbal autopsy: current practices and challenges. Bull World Health Organ. 2006;84(3):239–45.
- Serina P, Riley I, Hernandez B, Flaxman AD, Praveen D, Tallo V, et al. The paradox of verbal autopsy in cause of death assignment: symptom question unreliability but predictive accuracy. Popul Health Metr. 2016 [cited 2020

Apr 17];14(1). Available from: http://pophealthmetrics.biomedcentral.com/ articles/10.1186/s12963-016-0104-2.

- 20. King G, Lu Y. Verbal autopsy methods with multiple causes of death. Stat Sci. 2008;23(1):78–91.
- 21. Thomas L-M, D'Ambruoso L, Balabanova D. Verbal autopsy in health policy and systems: a literature review. BMJ Glob Health. 2018;3(2):e000639.
- 22. Bassat Q, Ordi J, Vila J, Ismail MR, Carrilho C, Lacerda M, et al. Development of a post-mortem procedure to reduce the uncertainty regarding causes of death in developing countries. Lancet Glob Health. 2013;1(3):e125–6.
- Castillo P, Ussene E, Ismail MR, Jordao D, Lovane L, Carrilho C, et al. Pathological methods applied to the investigation of causes of death in developing countries: minimally invasive autopsy approach. PLoS One. 2015; 10(6):e0132057.
- 24. Gillio-Meina C, Zielke HR, Fraser DD. Translational research in pediatrics IV: solid tissue collection and processing. Pediatrics. 2016;137(1):e20150490.
- Weustink AC, Hunink MGM, van Dijke CF, Renken NS, Krestin GP, Oosterhuis JW. Minimally invasive autopsy: an alternative to conventional autopsy? Radiology. 2009;250(3):897–904.
- Byass P. Minimally invasive autopsy: a new paradigm for understanding Global Health? PLoS Med. 2016;13(11):e1002173.
- Ben-Sasi K, Chitty LS, Franck LS, Thayyil S, Judge-Kronis L, Taylor AM, et al. Acceptability of a minimally invasive perinatal/paediatric autopsy: healthcare professionals' views and implications for practice. Prenat Diagn. 2013;33(4):307–12.
- Ngwenya N, Coplan D, Nzenze S, Myburgh N, Madhi S. Community acceptability of minimally invasive autopsy (MIA) in children under five years of age in Soweto, South Africa. Anthropol South Afr. 2017;40(2):108–21.
- 29. Feroz A, Ibrahim MN, McClure EM, Ali AS, Tikmani SS, Reza S, et al. Perceptions of parents and religious leaders regarding minimal invasive tissue sampling to identify the cause of death in stillbirths and neonates: results from a qualitative study. Reprod Health. 2019;16(1):53.
- Maixenchs M, Anselmo R, Martínez Pérez G, Oruko K, Agnandji ST, Angoissa Minsoko PC, et al. Socio-anthropological methods to study the feasibility and acceptability of the minimally invasive autopsy from the perspective of local communities: lessons learnt from a large multi-Centre study. Glob Health Action. 2019;12(1):1559496.
- Lewis C, Latif Z, Hill M, Riddington M, Lakhanpaul M, Arthurs OJ, et al. "We might get a lot more families who will agree": Muslim and Jewish perspectives on less invasive perinatal and paediatric autopsy. Idler E, editor. PLOS ONE. 2018;13(8):e0202023.
- Kang X, Cos T, Guizani M, Cannie MM, Segers V, Jani JC. Parental acceptance of minimally invasive fetal and neonatal autopsy compared with conventional autopsy: parental acceptance of MIA in fetuses and neonates. Prenat Diagn. 2014;34(11):1106–10.
- Lewis C, Hill M, Arthurs OJ, Hutchinson JC, Chitty LS, Sebire N. Health professionals' and coroners' views on less invasive perinatal and paediatric autopsy: a qualitative study. Arch Dis Child. 2018;103(6):572–8.
- 34. Feroz A, Ali AS, Ibrahim MN, McClure EM, Tikmani SS, Reza S, et al. Perceptions of health professionals regarding minimally invasive tissue sampling (MITS) to identify the cause of death in stillbirths and neonates: results from a qualitative study. Matern Health Neonatol Perinatol. 2019;5(1):17.
- Maixenchs M, Anselmo R, Sanz A, Castillo P, Macete E, Carrilho C, et al. Healthcare providers' views and perceptions on post-mortem procedures for cause of death determination in Southern Mozambique. Heazell A, editor. PLOS ONE. 2018;13(7):e0200058.
- 36. Blevins J, O'Mara Sage E, Kone A, Maixenchs M, Raghunathan PL, Guilaze RA, et al. Using Participatory Workshops to Assess Alignment or Tension in the Community for Minimally Invasive Tissue Sampling Prior to Start of Child Mortality Surveillance: Lessons From 5 Sites Across the CHAMPS Network. Clin Infect Dis. 2019;69(Supplement\_4):S280–90.
- 37. Das MK, Arora NK, Rasaily R, Chellani H, Gaikwad H, Banke K. Exploring family, community and healthcare provider perceptions and acceptability for minimal invasive tissue sampling to identify the cause of death in under-five deaths and stillbirths in North India: a qualitative study protocol. Reprod Health. 2019;16(1):3.
- Bunei M, Muturi P, Otiato F, Njuguna HN, Emukule GO, Otieno NA, et al. Factors influencing acceptance of post-mortem examination of children at a tertiary Care Hospital in Nairobi, Kenya. Ann Glob Health. 2019;85(1):95.
- Castillo P, Ussene E, Ismail MR, Jordao D, Lovane L, Carrilho C, et al. Pathological Methods Applied to the Investigation of Causes of Death in Developing Countries: Minimally Invasive Autopsy Approach. Cappello F, editor. PLOS ONE. 2015;10(6):e0132057.

- Ben-Sasi K, Chitty LS, Franck LS, et al. Acceptability of a minimally invasive perinatal/paediatric autopsy: healthcare professionals' views and implications for practice. Prenat Diagn. 2013;33(4):307–12. https://doi.org/10.1002/pd. 4077.
- Horey D, Flenady V, Conway L, McLeod E, Yee KT. Decision influences and aftermath: parents, stillbirth and autopsy. Health Expect Int J Public Particip Health Care Health Policy. 2014;17(4):534–44.
- Heazell AEP, McLaughlin M-J, Schmidt EB, Cox P, Flenady V, Khong TY, et al. A difficult conversation? The views and experiences of parents and professionals on the consent process for perinatal postmortem after stillbirth. BJOG Int J Obstet Gynaecol. 2012;119(8):987–97.

# **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

#### Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

#### At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

